Author's response to reviews

Title: The burden of disease profile of residents of Nairobi's slums: Results from a Demographic Surveillance System

Authors:

  Catherine Kyobutungi (ckyobutungi@aphrc.org)
  Abdhalah K Ziraba (aziraba@aphrc.org)
  Alex Ezeh (aezeh@aphrc.org)
  Yazoume Ye (yyazoume@aphrc.org)

Version: 3 Date: 8 February 2008

Author's response to reviews:

The following revisions have been made:

1) Figures 2 and 3 have been combined into one as suggested by the editors. This entailed omitting the person years from figure 2 and adding the numbers from the original figure 3 on a secondary y-axis.

2) the text in the results section has been changed only slightly to reflect the above changes

3) the figure legends have also been changed accordingly